Author's response to reviews

Title: The impact of a cancer diagnosis on weight change: findings from prospective, population-based cohorts in the UK and the US

Authors:

Sarah E Jackson (s.e.jackson@ucl.ac.uk)
Kate Williams (k.williams.11@ucl.ac.uk)
Andrew Steptoe (a.steptoe@ucl.ac.uk)
Jane Wardle (j.wardle@ucl.ac.uk)

Version: 5 Date: 17 November 2014

Author's response to reviews: see over
Dear Dr Keku,

**The impact of a cancer diagnosis on weight change: findings from prospective, population-based cohorts in the UK and the US [MS 9142471431339054]**

We are delighted to hear that our paper will be published in *BMC Cancer*. We have made the following revisions to the manuscript:

1. **Please can you clarify whether the data that you have used from these two studies are publicly available. If not, please clarify who granted permission to access data from these two studies.**

   - **Response:** The data for both studies are publicly available. We have added a sentence to page 4 of the manuscript detailing web addresses where data can be downloaded.

Yours sincerely,

Jane Wardle, MA, PhD, FMedSci
Professor of Clinical Psychology
Director: Health Behaviour Research Centre